See discussions, stats, and author profiles for this publication at: https://www.researchgate.net/publication/343444852

Analysis of Influence of Segmentation, Features, and Classification in sEMG Processing: A Case Study of Recognition of Brazilian Sign Language Alphabet

Article in Sensors · August 2020



Desenvolvimento de dispositivos de captura de sinais eletromiográficos da face View project



Article

Analysis of Influence of Segmentation, Features, and Classification in sEMG Processing: A Case Study of Recognition of Brazilian Sign Language Alphabet

José Jair Alves Mendes Junior¹, Melissa La Banca Freitas², Daniel Prado Campos³, Felipe Adalberto Farinelli¹, Sergio Luiz Stevan Jr.^{2,*} and Sérgio Francisco Pichorim¹

- ¹ Graduate Program in Electrical Engineering and Industrial Informatics (CPGEI), Federal University of Technology–Paraná (UTFPR), Curitiba (PR) 80230-901, Brazil; josej@alunos.utfpr.edu.br (J.J.A.M.J.); felipe.farinelli@outlook.com (F.A.F.); pichorim@utfpr.edu.br (S.F.P.)
- ² Graduate Program in Electrical Engineering (PPGEE), Federal University of Technology–Paraná (UTFPR), Ponta Grossa (PR) 84017-220, Brazil; melissa.1995@alunos.utfpr.edu.br
- ³ Graduate Program in Biomedical Engineering (PPGEB), Federal University of Technology–Paraná (UTFPR), Curitiba (PR) 80230-901, Brazil; danielcampos@utfpr.edu.br
- * Correspondence: sstevanjr@utfpr.edu.br

Received: 9 July 2020; Accepted: 1 August 2020; Published: 5 August 2020



Abstract: Sign Language recognition systems aid communication among deaf people, hearing impaired people, and speakers. One of the types of signals that has seen increased studies and that can be used as input for these systems is surface electromyography (sEMG). This work presents the recognition of a set of alphabet gestures from Brazilian Sign Language (Libras) using sEMG acquired from an armband. Only sEMG signals were used as input. Signals from 12 subjects were acquired using a MyoTM armband for the 26 signs of the Libras alphabet. Additionally, as the sEMG has several signal processing parameters, the influence of segmentation, feature extraction, and classification was considered at each step of the pattern recognition. In segmentation, window length and the presence of four levels of overlap rates were analyzed, as well as the contribution of each feature, the literature feature sets, and new feature sets proposed for different classifiers. We found that the overlap rate had a high influence on this task. Accuracies in the order of 99% were achieved for the following factors: segments of 1.75 s with a 12.5% overlap rate; the proposed set of four features; and random forest (RF) classifiers.

Keywords: sign language; surface electromyography; wearable device; signal segmentation; feature extraction; pattern recognition; machine learning

1. Introduction

Sign Language recognition (SLR) systems have an important task in aiding the communication between deaf people and the hearing impaired. These systems recognize Sign Language, a structured set of manual gestures that are used by the deaf and hearing impaired population as a communication system [1]. It is known that social communication is one of the barriers that people with hearing disabilities face, especially when they need to establish communication with people that do not know Sign Language (SL) [2]. In many cases, even if sign language is an official (native) language that allows the cognitive development of an individual [3], the communication between deaf and speaking people presents some failures when the speakers do not have enough knowledge of SL [4], which represents the majority of the speakers. Thus, an SLR system translates the SL signs to text or speech to ease the communication between people with deafness and the speakers [5,6].



There are two main technologies of SLRs: ones based on images (which use cameras to capture the signals), and ones based on sensors (which use sensors placed on the upper limbs or wearable devices, such as wristbands, gloves, and armbands) [1,7]. Although the techniques with an image approach have been widely used, they are highly dependent on environmental factors, such as fixed placement of the camera, brightness and contrast. On the other hand, sensor-based SLR systems use motion devices, inertial sensors, and electrical biosignals, such as surface electromyography (sEMG), as input. These devices are a relevant proposal because they have more robustness since the signal can be acquired in any environment. In particular, sEMG signals have seen recent advances due to the development of wearable devices [8] and data processing techniques [9] for gesture recognition [10]. Wearable sEMG sensors-based SLR/gesture recognition systems have become more popular in recent years because of they are low cost, privacy nonintrusive, and have a ubiquitous sensing ability compared to vision-based approaches [11]. Nevertheless, it is necessary to mention that this technique has some limitations for SLR systems. Wearing a device can limit the action of a subject during the performance of a sign (especially if the device causes some discomfort for the subject), which restricts some applications for laboratory or controlled environments [1], and sEMG can generate data with a noisy aspect [12], which are difficult to be visually interpreted. Furthermore, in SLR applications, some signs combine hand movements and facial expressions, requiring the use of complementary devices to both acquire facial expressions and perform signal recognition.

sEMG is a technique used to acquire, process, and record electrical activity from skeletal muscles [13]. This technique uses surface electrodes placed on the skin. The obtained signal is a spatial-temporal sum of the action potentials that are generated in the muscles during their contraction, which is commanded by the nervous system [14]. The sEMG is useful for SLR, which is considered a gesture recognition application, because in contrast to other kinds of sensors (such as inertial and flexible resistors), the extracted signal presents data related to movement execution and its respective intention: one of the motives for its use is in applications for amputees [12]. Multichannel sEMG signals, recorded by a set of sEMG sensors placed on the arms, contain rich muscle information associated with different sign gestures [15].

During the processing of sEMG signals, segmentation, feature extraction, and classification stages are a key factor to guarantee high hit rates in pattern recognition [16]. However, these stages have several parameters that need great attention. In the segmentation, these include the muscle activation method, the window size, and the overlapping influence in the classification process [17]. The choice of feature sets to be extracted is the target for various scientific studies. New combination sets have been presented in the literature, seeking to improve accuracy and robustness for sEMG gesture recognition [18–20]. Several classification techniques are used, especially based on supervised machine learning algorithms [1], such as support vector machines (SVM) and artificial neural networks (ANN) [17,21]. For different applications, different features and different classifiers may exhibit significantly different performance. It is advisable to try different approaches to determine the best one.

Thus, this work presents the pattern recognition process used for the development of an SLR for Brazilian Sign Language (Libras) alphabet using sEMG signals as input. Libras was chosen due to it having similar gestures to other SLs and, in Brazil, this SL is recognized as an official language. Even though sEMG was explored for gesture recognition in other SLs around the world, such as American [11,22], Chinese [23,24], and Greek [10], there are few studies that apply for Libras. Regarding the sEMG signals, the influence of each processing stage is explored: the segmentation and windowing of sEMG signals, feature extraction, and their relation with the classification. In this study, the armband acquisition approach (using the commercial MyoTM armband) was considered due to the recent advances in sEMG pattern recognition using these devices and their popularization in similar researches. These recognition systems enable real-time operation and, at the same time, are relatively simple to implement.

The hypothesis that sEMG features extracted from sliding window segmentation could recognize Brazilian SL is evaluated in this work. At the same time, the parameters of a sliding window are analyzed to achieve the best classification performance. Additionally, as most previous works considered the fusion of IMU (Inertial Measurement Unit) and sEMG, the efficiency of sEMG signals alone remains unclear. In this way, this work also brings some results of this modality of sensor to Libras recognition. Besides that, the effects of some parameters in the recognition process are little explored. For example, what is the influence of segmentation parameters, such as the window length and the overlapped segments, in an SLR system? Alternatively, which features (or feature sets) could be used to improve a sEMG-based SLR systems? Moreover, which classifier (or classifiers) could be more suitable for an sEMG armband device for Libras recognition? These questions arise in the process of SLR recognition, and in some related works these questions are not fully answered [11,23,25–27]. Thus, a methodology is presented in this paper to evaluate how the sEMG processing parameters

influence classification systems for Brazilian Sign Language. This work is organized as follows: Section 2 presents the related works evolving sEMG for SLR; Section 3 presents the data acquisition device, the protocol to record the sEMG signals, and the methodology to process the signals; Section 4 presents the results obtained for each methodology stage; and Section 5 presents the main conclusions of this work.

2. Related Works

In the gesture recognition field, the sEMG signal is widely explored and applied as an input [1]. In recent years, the number of works that apply sEMG as input for SLRs significantly increased. The majority of these works focused on American (ASL) and Chinese (CSL) Sign Languages and used multi-channel acquisition (more than one sEMG channel). Moreover, it is common to employ inertial sensors for data fusion to ease the sEMG segmentation and to increase the accuracy of classification in dynamic gestures.

The first studies for SLR and sEMG were conducted to classify nine signals from ASL using 16 features and discriminant analysis (DA) [25]. Two channels were placed on the forearm muscles, allowing hit rates of about 97%. However, there no information was presented about the number of subjects and their characteristics. Afterward, 40 [28] and 80 [11] signals from ASL were recognized, with a high number of features and classifiers. Four channels (fixed on the forearm muscles) and inertial sensors fusion were able to classify the gestures for four subjects with a 96% hit rate with several sEMG features.

Many works aimed for the recognition of signals from CSL. Zhang et al. (2011) [24] applied hidden Markov models (HMMs) to classify 72 signals using five channels and accelerometers in two subjects. In 2015, 223 gestures acquired from five subjects were classified with accuracies from 92% to 96% using HMM models and the dynamic time warping (DTW) technique [29]. Random forests (RF) and HMM were employed to recognize 121 phonemes from CSL for five subjects using five channels and an accelerometer sensor [27]. Moreover, Zhuang et al. (2017) [23] demonstrated that the window size interferes with the classification process. At least 18 signals acquired from eight subjects were classified using linear discriminant analysis (LDA), reaching a 91% accuracy level [23].

Concerning other SLs, 60 gestures from Greek Sign Language (GSL) were classified using discriminant analysis and sEMG sample entropy with 93% hit rates [10]. In this study, five channels were placed on the forearm of three subjects and data from the accelerometer were used for classification. Moreover, 30 gestures in Korean Sign Language (KSL) were classified for six subjects using the commercial MyoTM armband [26]. The raw signals from the inertial sensors were inserted into a convolutional neural network (CNN), achieving a 98% accuracy level. Other works presented the recognition for Thai Sign Language (TSL) [30] and Indonesian Sign Language (ISL) [31,32]; however, in their methodologies, some aspects were not explained and described, such as the number of subjects from which the data were collected.

Regarding Libras, it is necessary to point out that image-approach was the tool for several studies for SLRs [33–35]. The sensor approach is based on instrumented gloves, and sEMG wearable devices are a field that is being explored in recent works, achieving high accuracies and hit rates [36,37]. There

4 of 25

is a limited number of works combining sEMG and Libras, and their focus is on alphabet classification. Abreu et al. (2016) [38] used the MyoTM armband to classify 20 static letters with an SVM classifier. The accuracy for each letter was presented instead of the overall accuracy; however, the average of hit rates in the test stage was about 41%. This work demonstrated that it is suitable to classify Libras from sEMG signals, but the authors pointed out that limitations in processing and acquisition hindered achieving a high performance. Posteriorly, the same process was repeated for 20 static letters by Mendes Junior et al. (2019) [39]. From one subject, seven time-domain and five frequency-domain features were extracted from the sEMG signals, and a multi-layer perceptron (MLP) neural network was used as the classifier. In this approach, the overall accuracy was above 80% during the test step. In this work, it the errors that can occur during the classification of these gestures were verified, such as the gesture "a" misclassified as "s", "n" as "p", and "g" as "w". In sEMG applications, the evaluation of the influence of acquisition and processing steps, such as number of channels, segmentation methods, features extraction, and type of classifiers, is missing.

Some aspects can be raised concerning the characteristics of these works. The first is that this application can reach high hit rates; however, the number of subjects used in these studies was low (less than 10 subjects), which can explain the value of the accuracies. In data acquisition, the armband approach was been considered (e.g., MyoTM armband) instead of the fixed electrode placement. In segmentation, the overlapped segmentation was applied in some works, but its variation was not evaluated. Even knowing that this parameter (together with window length) affects the pattern recognition process [17,40], the effect of overlapped segments. During the classification, the contribution for each feature or the use of specific feature sets and their influence on accuracy were not evaluated. The most repeated features were mean absolute value (MAV) and autoregression (AR), but this does not explain why these features were used rather than others available in sEMG processing. Regarding the number of classifiers, different techniques were used, but the behavior for more than four techniques was not presented.

Other recent studies presented frameworks and techniques that have been developed to solve the SLR problem, focusing on signal processing for sensors with different natures. For example, a leap motion sensor was used with a hidden Markov layer for pattern recognition of 24 gestures of ASL [41] and instrumented gloves with flex [42] and/or inertial sensors [43] with machine learning techniques reached accuracies ranging from 86% to 98%. A combination of images and deep learning techniques demonstrated high performance for this task, such as for Persian Sign Language [44] and CSL [45]. This indicates that this topic has been developed recently. The most prominent similarity shared by studies with the sEMG studies was signal processing, which is a recurring demand for this application.

3. Materials and Methods

In this section, the data acquisition device, the experimental protocol, the processing technique, and the classification methods are presented.

3.1. Data Acquisition

The Ethical Committee for Research in Humans of the Federal University of Technology—Paraná (CAAE 89638918.0.0000.5547) approved the data acquisition. The chosen gesture set was the Libras alphabet, composed of 26 gestures (20 static gestures and six dynamic gestures) presented in Figure 1. It is a closed set of gestures, which is used by Libras speakers to spell words, acronyms, names, and facilitate the construction of other signals (e.g., the signal related to the name of a person or technical or specialized vocabulary) [34]. In addition, these signals use different hand configurations and some movements, which are suitable for gesture recognition applications.



Figure 1. Alphabet set from Brazilian Sign Language (Libras). The set is composed of 26 letters, including 20 static gestures and six dynamic gestures.

The commercial MyoTM armband was chosen for data acquisition, and is presented in Figure 2a. This device is widely used in gesture recognition research [46,47]. MyoTM has eight equidistant channels, with 200 samples/s, 8-bit resolution from ADC (Analog to Digital Converter), and wireless communication via Bluetooth [48]. Signal acquisition is made through dry stainless-steel electrodes and there is a 3-axis sensor unit with an accelerometer, gyroscope, and magnetometer. MyoTM was placed on the right forearm of subjects (Figure 2b), which is the orientation needed to place channel 3 on the *flexor carpi ulnaris* muscle. All the subjects were right-handed and the gestures were performed only with the right hand.



Figure 2. (a) Commercial MyoTM device used for data acquisition with the numbers of its respective channels. (b) Placement of the armband on the subjects' forearm. The reference (channel 3) was placed on the *flexor carpi ulnaris* to make all acquisitions uniform.

Signals from 12 subjects (10 men and two women), with an average age of 23.2 ± 2.4 years, were acquired (for more information, see Appendix A). The subjects were healthy, without myopathies or muscle diseases in the upper limbs. Figure 3 presents the signal acquisition process. A graphical interface was developed to inform subjects what gesture should be performed and the next gesture in the sequence. When the subject was required to perform a gesture, a light indicator turned on. The indicator turned off when the subject was required to rest their forearm muscles. In each change, a buzzer sounded. The time interval for each gesture was defined with a frequency of 45 beats per minute (bpm), equivalent to 1.3 s between performing each gesture and rest state. In each trial, the subjects performed the 26 gestures; this sequence was repeated 20 times for each subject.



Figure 3. Sequence for sEMG signals acquisition. When the acquisition started, a light and a buzzer indicated that the subject should perform the gesture. After the 1.3 s, the light indicator turned off and the buzzer indicated to the subject to rest. After 1.3 s, both indicators turned on, repeating this process until the acquisition was completed.

3.2. Signal Processing

The sEMG processing for gesture classification can be divided into three main stages: segmentation, feature extraction, and pattern recognition.

3.2.1. Segmentation

Segmentation is responsible for detecting and windowing the sEMG signal. In this work, the detection of the gesture was made using the double threshold onset technique. The onset was based on the energy for the eight channels, calculated as described in [27]. The threshold was defined as 10% of the maximum energy for each trial compared with a time interval of 150 ms to avoid false triggers and noises [49].

In segmentation, two main parameters were evaluated: the window length and the overlap rate. The window range was defined from 0.25 to 2.25 s, with steps of 0.25 s. These values were defined taking into account the delay for sEMG processing [50,51] and the time at which the gestures were performed in the acquisition. To analyze the overlapping, the parameter overlap fraction (ϕ) was calculated by:

$$\phi = k/W, \tag{1}$$

where *W* is the window length of each segment and *k* is the time step from one segment to the next. This means that an overlap fraction of 1 (100%) represents a disruptive window, 0.5 (50%) a half-step overlapping window, 0.25 (25%) a quarter-step overlapping window, and 0.125 (12.5%) an eighth-step overlapping window. Four parameters were analyzed during overlapping: 100% (a disruptive window), 50%, 25%, and 12.5%. This method is more effective than the use of a fixed step for overlap signals for evaluation due to having different window segments to analyze. The overlapping process may also include information that is present in the adjacent signal window; therefore, it works as a redundant extraction to ensure that the useful information is contained inside the analysis segment [51].

3.2.2. Feature Extraction

Feature extraction is the stage where the dimensionality from the segments is reduced into useful information for classifiers, especially for the sEMG signals that have a stochastic nature [52]. The features analyzed in this work are presented in Table 1 with their respective parameters for implementation. In total, 33 features were extracted and evaluated: 24 features from the time domain and nine from the frequency domain. These features were chosen due to their use in other feature sets in the literature [19,53–55], which present their math expressions. Frequency features were extracted after the signals passed to a fast Fourier transform in each segment with 2000 points of resolution. Some features needed parameterization and their values were based on studies that applied them for gesture recognition. For auto-regressive and cepstral coefficients, the fourth order was chosen [53,56];

in the histogram feature, nine bins were extracted [57]. The second-moment was used in L-Scale [55]; the third order was chosen for V-order [53]; and a threshold value of 10^{-2} was defined for the Myopulse percentage rate, Willison amplitude, sign slope change, and zero crossing features [58]. A 0.2 σ tolerance (standard deviation) and two dimensions were chosen for sample entropy [59]; the band of frequency was split at the average for the frequency ratio [53].

Domain	Feature	Feature Name	Parameters	
	AR4	4th-order Autoregressive Coefficients	_	
	CEPS	Cepstral Coefficients	4th-order	
	DASDV	Difference Absolute Standard Deviation Value	-	
	HIST	Histogram	9 Bins	
	IEMG	Integral of EMG	-	
	LOGDEC	Log Detector	-	
	LS	L-Scale	Two Moment	
	MAV	Mean Absolute Value	-	
Timo	MAV1 and MAV2	MAV1 and MAV2 Modified Mean Absolute Value		
Time	MFL Maximum Fractal Length		-	
	MSR	Mean Square Root	-	
	MYOP Myopulse percentage rate		Threshold = 10^{-2}	
	RMS	Root Mean Square	-	
	SampEn	Sample Entropy	Dimension = 2 $r = 0.2 \sigma$	
	SSC	Sign Slope Change	Slope threshold = 10^{-4}	
	TM3, TM4, and TM5	Absolute Value of 3rd, 4th and 5th Moments	-	
	VAREMG Variance		-	
	VORDER V-Order		3 Order	
	WAMP Willison Amplitude		Threshold = 10^{-2}	
	WL Waveform Length		-	
	ZC	Zero Crossing	Amplitude threshold = 10^{-2}	
Frequency	FR	Frequency Ratio	Low frequencies = 10–50 Hz High frequencies = 51–100 Hz	
	MDF	Median Frequency	-	
	MNF	Mean Frequency	-	
	MNP	Mean Power Spectrum	-	
	PKF	Peak Frequency	-	
	SM1, SM2, and SM3	Spectral Momentum	-	
	TTP	Total Power Spectrum	-	

Table 1. Time-domain and frequency-domain features extracted from this work.

The performance of the features in the classification process was also compared with feature sets presented in the literature [54–56]. The features in Table 1 were analyzed individually and organized in feature sets based on their accuracy in the pattern recognition process. Some groups were tested: a set with the most repeated features found in related works (Section 2), a set of the best features with high performance individually, a set with the best time-domain features, a set with the best frequency-domain features, and a reduced set of features with relevance to accuracy and statistical analysis. The features sets are summarized in Table 2.

Feature Set	Features
G1	Hudgins et al. (1993) set: MAV, WL, ZC, and SSC [54];
G2	Liu et al. (2007) set: AR4 and HIST [56]
G3	Most repeated features in Section 2 (Related Works): MAV and AR4;
G4	TD4 set: MFL, MSR, WAMP, and LS [55];
G5	TD9 set: LS, MFL, MSF, WAMP, ZC, RMS, IAV, DASDV, and VAREMG [55]
G6	Best features performed individually
G7	Best time-domain features in G6
G8	Best frequency-domain features in G6
G9	Reduced feature set from G6 with relevance in accuracy based on statistical analysis

Table 2. Feature Sets organized and applied in this work.

3.2.3. Classification

In the pattern recognition stage, classifiers label the gestures based on the inputs. In this work, the following classifiers were applied: *k*-nearest neighbor (KNN), linear discriminant analysis (LDA), probabilistic naïve Bayes (NB), quadratic discriminant analysis (QDA), extreme learning machine (ELM) and multi-layer perceptron (MLP) neural networks, random forest (RF), and support vector machines based on linear discrimination (SVMLin) and Gaussian kernel-type based on radial basis function (SVMRBF). These classifiers were selected based on the classifiers listed in related works (Section 2) and presented high accuracies in gesture classification [27,53,60,61]. The *k*-fold cross-validation method was used, using 10 folds.

Regarding the classifiers, some of them need parameterization, which was chosen using the cross-validation procedure. One nearest neighbor was used in KNN. The hyperbolical function was used in the neural networks and the number of neurons in the hidden layer for the ELM was 1000 and in MLP 30. In random forests, 30 trees were selected. The SVMs were built based on the LibSVM library [62]. In SVMLin, the *C* parameter was defined as 100; in SVMRBF *C* was programmed as 10 and the size of Gaussian functions was defined as 1. Table 3 summarizes their parameterization.

Classifian	Paramatore
Classifier	ratailleters
k-Nearest Neighbor (KNN)	1–nearest neighbor
Linear Discriminant Analysis (LDA)	-
Naïve Bayes (NB)	Normal distribution
Multi-Layer Perceptron (MLP)	30 neurons in hidden layer
Quadratic Discriminant Analysis (QDA)	-
Random Forest (RF)	30 trees
Extreme Learning Machine (ELM)	1000 neurons in the hidden layer
Support Vector Machine with Linear Discrimination (SVMLin)	C = 100
Support Vector Machine with Radial Basis Function Discrimination (SVMRBF)	C = 10 and Gaussian size = 1

Table 3. Classifiers applied in this work and their parameterization.

3.3. Methodology to Data Processing

A methodology was developed to decrease the computational effort to find the parameters of this SLR. Figure 4 presents the steps performed for data processing. Firstly, the data were segmented in a fixed window length and an overlap rate with 100%, which means that the segments are windows without overlap (disruptive segments). These values in segmentation, and especially the window length and no presence of overlapping, were chosen due to high accuracies in similar works. Thus, the analysis was performed in the features extracted in Table 1.

The features were classified individually and the best features and the classifiers with high accuracies were selected. These features compounded the aforementioned groups from G6 to G8.



Figure 4. Steps in data processing methodology.

The second stage was the analysis for the feature sets (G1 to G8), particularly the features presented in the literature (G1 to G5) and the features in the previous stage (G6 to G8). The best feature sets were selected for segmentation analysis, which was the third stage. The accuracies for ranges of window length and overlap rate values were calculated for the selected classifiers, searching for the best combination of these two parameters. As aforementioned in Section 3.2.3, the pattern recognition was performed using k-fold cross-validation (with k equals to 10) for the entire dataset in the analysis from stages 1 to 3. Thus, this method was chosen to verify the influence of the parameters for the entire dataset, without taking into account the difference between the subjects.

The last stage was the analysis of accuracy for each subject. Two methods were used to evaluate each subject. Firstly, the data from each subject were evaluated separately; in each training and test samples were from the same subject. Afterwards, the training and test sets were separated based on the acquisition trials, with k-fold cross-validation based on the acquisition trials (Figure 3). For example, *k* number of trials were removed for training data for each subject. The test set was composed of these *k*-trials samples and tested individually for each subject. This analysis can show how the pattern recognition system behaves for samples from the subjects with a previous calibration (provided for the training samples) while also considering all the full datasets and involving all subjects. As it is based on a calibration that occurs in a wearable device for sEMG (the MyoTM) [63], it provides a generalist vision for the pattern recognition process. In both analyses, the data were partitioned using a 10-fold cross validation method.

4. Results and Discussion

The results are presented following the stages described in the methodology (Figure 4). Firstly, the accuracy for each feature and classifier are presented individually, followed by the analysis of feature sets. Posteriorly, the influence of segmentation is considered and the tests for each subject are exhibited. Finally, the results are compared with those presented in the literature (related works).

4.1. Feature and Classifier Separately

The first step consisted in the evaluation of the behavior of each feature individually. Figure 5 presents the accuracies for the features (presented in Table 1) extracted individually for all the classifiers. In this analysis, the window length and overlap rate were set as 1 s and 100% (a disruptive window without overlap), respectively. It can be noted that there are features that have high accuracies comparing to others, which is common in sEMG processing. Moreover, the use of an isolated feature cannot provide an acceptable level of accuracy. Some features, such as in the frequency domain (e.g., Mean Frequency, Median Frequency, and Frequency Ratio) and with complex information (e.g., 4-th order Autoregressive Coefficients and Sample Entropy) had accuracies close to prior probability (1/26), which are not being indicated in this application. Features with a high correlation of amplitude levels, such as in the time domain, which presented performances of up to 50% for some classifiers, were indicated more in this case. This reinforces a feature set needs to be built.



Figure 5. Accuracies for each extracted feature for all the classifiers considered in this work. Their abbreviations are listed in Table 1 and in Abbreviations section. As segmentation parameters, a window of 1 s and an overlap fraction of 1 (100%, a disruptive window without overlapping) were considered. The mean for each feature is presented by the dashed lines. The features with high accuracies were used for the creation of the feature sets G6 to G9.

A mean for the performance of all classifiers was extracted (presented by the dashed lines) to identify the best features and classifiers in this scenario. Regarding the classifiers, one can note that four of them had accuracies of up to the mean: KNN, ELM, RF, and SVMRBF. This behavior can be noted in similar works for sEMG signals that were compared with different classifiers [53,64]. Because of these values, these four classifiers were analyzed more comprehensively, aiming to reduce the computational effort.

In decreasing order of hit rates, the features with better performance were MFL, MNP, TTP, RMS, SM1, LS, DASDV, SM2, SM3, IEMG, MAV, WL, MSR, MAV1, HIST, VAREMG, and MAV2. Based on these features, the feature sets G6 to G9 were organized as previously mentioned. G6 was composed of all 17 features. G7 was composed of the best 12 features in the time domain: MFL, RMS, LS, DASDV, IEMG, MAV, WL, MSR, MAV1, HIST, VAREMG, and MAV2. G8 was composed of MNP, SM1, SM2, SM3, and TTP.

The feature set G9 was composed of the four best features from G6 that demonstrated high classification accuracy with a reduced number of features. The reduction was based on the statistical dependency of the distribution after the pattern recognition process (using a Friedman test and Tukey's post-hoc test for multi-comparison, p > 0.05) and the following features were selected: MFL, MNP, TTP, and RMS. This process is presented in Figure 6, composed of the features with the highest accuracies in G6 and the best four classifiers (ELM, KNN, RF, and SVMRBF). Figure 6a presents the accuracy rates obtained with the increase of each feature for the classification. It can be noted that when increasing the number of features, their medians stabilized for four features (up to Root Mean Square, RMS). After this, there was no significant gain and the coverage of distribution was more concentrated. In Figure 6b, it can be noted that the distribution of these four features overlaps for almost all distributions (except the first). The four features presented equivalent distribution when more features were inserted into the classification (Figure 6b), besides that the range of their distribution (denoted by the whiskers of Figure 6a) covered the values when more features were introduced into the classification. With these definitions, the next analysis was the influence of the feature sets.



Figure 6. The selection process of the features from G6 to G9. The boxplot in (**a**) presents the distribution of accuracies increasing the number of features of G6 (from the highest to the smallest hit rate obtained separately) for the best classifiers (ELM, KNN, RF, and SVMRBF). The bars represent the 25th and 75th percentiles; the whiskers represent approximately 99.3% of the coverage of distribution, and the central mark represent the medians. (**b**) Distributions for the Tukey post-hoc from Friedman statistical test to evaluate the distribution of results with a confidence interval of 95% (*p*-value < 0.05) when increasing of the number of features in the classification process. The highlight indicates the distribution that is related to high accuracy obtained with a reduced number of features in (**a**) that corresponds statistically with the several distributions as demonstrated in (**b**).

Figure 7 presents the classification performance for the feature sets from the literature (G1 to G5) and the features with high hit rates in the previous section (G6–G9). The KNN, ELM, RF, and SVMRBF classifiers were considered in this analysis due to their high accuracies in the individual feature analysis. It can be noted that the accuracies show a slight increase using feature sets compared to the use of individual features, with means ranging from 40% to 60%. The best classifier for the feature sets alternated between RF and SVMRBF.



Figure 7. (a) Accuracies for the selected features sets (G1 to G9) for the classifiers that presented the highest accuracies in the previous analysis. (b) Distributions for the Tukey post-hoc from Friedman statistical test to evaluate the distribution of results with a confidence interval of 95% (*p*-value < 0.05). Two sets of ranks can be observed, separating G1 to G3 and G4 to G9. One can note that the feature sets G4, G5, G6, G8, and G9 have similar distributions.

The classical feature sets (G1 and G2) and the most repeated features in related works (G3) presented the worst performance compared with TD4 (G4), TD9 (G5), and the features established in the previous section. This could be explained due to the presence of features that cannot have high separability for the sampling rate used in the MyoTM device (such as ZC and SSC) [55] or have complex features (such as AR4).

Even though the best accuracies were present from G4 to G9, it was necessary to verify that they belonged to the same distribution. The accuracy distributions were submitted to a Tukey post-hoc from Friedman test, as presented in Figure 7b. The ranks presented the distribution, and the same groups were analyzed with a confidence interval of 95% (*p*-value < 0.05). There were two main distributions. G1, G2, and G3, with the lowest accuracies, belonged to the same distribution. From the feature sets with the highest accuracies, G4, G5, G6, G8, and G9 belonged to the same distribution, which was statistically different from the other feature sets.

Comparing the feature sets, three of them were considered for the subsequent analysis: G4 (TD4), G8 (best in the frequency domain), and G9 (reduced set from the best features). G5 was discarded because it had three of the four features of G4, which had a similar distribution. Considering the number of features, G5 had nine features whilst G4 had only four. This represents a decrease in the time used to extract these features, considering an embedded application. The same criterion was used to discard group G6, which had 17 features whilst G8 and G9 had five and four features, respectively.

As the previous analysis consisted of fixed values for window length and overlap rate, the next step, shown in Figure 4, was the variation of segmentation parameters. Figure 8 presents the results obtained for the variation of the overlap and the window size for the classifiers (ELM, KNN, RF, and SVMRBF) and the feature sets with high performance (G4, G8, and G9). Each color represents a level of accuracy, as demonstrated by the color bar. Dark blue represents the lowest accuracy whilst light yellow represents the highest one.



Figure 8. Results for the segmentation analysis. The color bar indicates the accuracy for each classifier on the influence of feature sets and segmentation parameters. The high accuracies are concentrated for large window length and small overlap rate.

The first fact to note is that an increase in the number of elements in the segment also increased the hit rates. This fact occurred for all classifiers, feature sets, and overlap rates. As mentioned before, in works related to sEMG pattern recognition, the window length directly affected the accuracies [17,65,66]. However, the window size cannot be too extensive or large and thus compromise the whole application (involving the detection of sEMG and the final response of the actuation). The accuracies showed significant increases for segments longer than 1 s (above 200 datapoints in each segment). The inverse occurred for overlap rate: no overlap (100%) or a high rate resulted in low accuracies, from 60% to 70% for both classifiers and feature sets. It is notable how the overlap process influenced the accuracy in every scenario. Then, it is necessary to highlight that for window lengths in the order of 0.25 and 0.5 s, some combinations, such as G4 for KNN and RF, presented hit hates in the order of 80 to 90% for 25% and 12.5% overlaps. Even though this was the highest accuracy that the recognition process could reach, it is an expressive value.

ELM and the G8 feature set had the lowest accuracies. The other classifiers had similar results, both for overlap and window length, and G4 and G9 presented similar distributions. Still, a window length of 2.25 s had high accuracy; such a length needs to search for a window length with the same distribution with a lower number of segments in a window. An analysis in groups was performed, using a Friedman test and a Nemenyi post-hoc test (p > 0.05) and considering the groups containing the segmentation parameters. Based on their results, critical distance diagrams (CD) were performed and are presented in Figure 9. These diagrams list the groups with the best ranks on the left and distributions without statistical difference are connected with a solid line [67]. Figure 9a presents the comparison related to the classifiers and window length. It can be noted that segments of 1.75 s have similar distributions to 2.25 s and 2 s, even though the latter two present the best performance rates.

Figure 9b presents the analysis of classifiers and feature sets from the overlap rate. The overlap rate of 12.5% showed ranks placed differently than 25%; however, there were areas of overlap between them.



Figure 9. Critical distance diagram (CD) obtained from Friedman test and Nemenyi post-hoc test. The test was performed for (**a**) window length and (**b**) overlap rate variation. On the left, the best average rank performances are indicated. The critical difference attributes are denoted by the lines that connect the distributions. They indicate where there are no statistical differences. It can be noted that a window length of 1.75 s presented similar distributions to 2.25 and 2 s in several classifiers. Even the 12.5% overlap rate showed the best ranks, and some distributions were highly related to values of 25%.

Due to overlapping found in the groups of Figure 9, the distributions were analyzed separately. Table 4 presents the similar accuracies in the segmentation analysis for the observed similar parameters, such as window length for 1.25 to 2.25 s and overlap rate for 25% and 12.5%. It should be highlighted that using 2.25 s, the classifiers had 99% accuracy; for some classifiers, such as SVMRBF and KNN, one second less only decreased the accuracy by 1% for a 12.5% overlap rate.

T47* 1	Accuracy (%) G4					
Window	KNN Overlap Rate		RF Overlap Rate		SVMRBF Overlap Rate	
Length						
	25	12.5	25	12.5	25	12.5
1.25 s	92.5 ^{ab}	98 ^{ab}	89.3 ^{ab}	94.9 ^{ab}	94.2 ^{ab}	98.3 ^{abc}
1.5 s	95.1 ^{ab}	99 ^{abc}	93.8 ^{abc}	97.5 ^{abc}	96.4 ^{abc}	99.1 ^{abcd}
1.75 s	97.8 ^{abc}	99.7 ^{bcd}	97.2 ^{cd}	99 bcd	98 ^{cd}	99.5 ^{abcd}
2 s	99.8 ^{abceF}	99.9 ^{cdeF}	98.9 ^{cde}	99.6 ^{de}	99.3 ^{cde}	99.8 ^{bcde}
2.25 s	99.9 ^{ceF}	99.9 ^{cdeF}	99.9 ^{deF}	99.9 ^{deF}	99.8 ^{de}	99.9 ^{de}
			C	G9		
1.25 s	93.2 ^{ab}	97 ^{ab}	90.3 ^{ab}	95.3 ^{ab}	93.8 ^{ab}	97.7 ^{ab}
1.5 s	95.8 ^{abc}	98.8 ^{abcd}	94.3 ^{abc}	97.7 ^{abcd}	96.3 ^{abc}	99 ^{abc}
1.75 s	97.5 ^{bdc}	99.4 ^{bcde}	96.9 ^{bcd}	98.7 ^{bcde}	97.9 ^{bcd}	99.4 ^{bcd}
2 s	98.8 ^{cde}	99.5 ^{bcde}	99.2 ^{cde}	98.2 ^{bcde}	98.9 ^{cde}	99.6 ^{cde}
2.25 s	99.9 ^{deF}	99.9 ^{cdeF}	99.9 ^{deF}	99.9 ^{cdeF}	99.7 ^{de}	99.9 ^{de}

Table 4. Accuracies from the classifiers with similar accuracies in segmentation analysis (KNN, RF, and SVMRBF). The range of 1.25 to 2.25 s for window length and 25 and 12.5 overlap rates were considered due to the similar results reached in the segmentation influence analysis.

 a,b,c,d,e Equivalent statistical distribution for each window length in its respective column (same overlap rate) from Tukey post-hoc from Friedman test (p > 0.05): 1.25 (a), 1.5 (b), 1.75 (c), 2 (d), and 2.25 (e) seconds. ^F Equivalent statistical distribution for each 25 and 12.5 overlap rate pair in its respective row according to Wilcoxon test (p > 0.05).

Aiming to verify if they belong to the same distributions, window length data (each column) were submitted to a Tukey post-hoc from Friedman test and the overlap data (row) were submitted to a Wilcoxon test. These analyses are also presented in Table 4 with the following indices: a (1.25 s), b (1.5 s), c (1.75 s), d (2 s), and e (2.25 s) are window length and * indicates the same distribution for overlap rate. For example, when a value is marked with (a), it means that its distribution is statistically significant for 1.25 s. The same process was performed for overlap rate analysis: when the value is marked with (*), it means that 25 and 12.5 have the same distribution.

Searching a segment with distributions equivalent to the obtained results for 2.25 s with a lower number of segments, the similar distribution found was for 1.75 s. Regarding the overlap rate, one can note that there are still similar values of accuracies; the distributions presented statistical differences for the interval of 5% in the Wilcoxon test. Separately, the 25% and 12.5% overlapping ratios had different distributions in several cases, which means that these parameters can be investigated in the individual subject analysis.

4.4. Analysis of Subjects

After the analysis of the subjects, it is necessary to attempt using the feature sets. One can note that G4 and G9 did not present significant differences between them, which was statistically verified. To choose one feature set for the subject analysis, the computation time was measured for both feature sets. Even though G9 has frequency-domain features (which means calculating a frequency transform), it is 20% faster than G4. It should be explained due to the features present in G4, it has more calculation complexity than G9 (e.g., as LS and MFL) that can represent a computational cost in embedded applications.

Considering G9 and the aforementioned segmentation parameters, the analysis started training and testing with the data from each subject. Figure 10 presents the boxplot distribution for the hit rates obtained for each subject from two perspectives: for overlap rate (a) and for the classifier (b).



Figure 10. Distribution of accuracies considering the data for each subject individually in the test and training steps. (a) Distributions concerning overlap rate and (b) distributions concerning classifiers. The bars represent the 25th and 75th percentiles; the whiskers represent approximately 99.3% the coverage of distribution, and the crosses (+) are the outliers.

The same behavior regarding the influence of the overlap rate could be observed in all subjects, as presented in Figure 10a. The distributions were concentrated close to the medians, with a low rate of outliers. Some subjects had a high increase in accuracy changing the overlap of the segments. This can be noted in subject 3, which exhibited an accuracy improvement from 40% to 90%. For other subjects, such as 6 and 7, the gain in accuracy was around 20%, with a hit rate of close to 100%. On the other

hand, Figure 10b shows that some classifiers had more concentrated distributions, independent of the subject. This was the case with the RF and SVMRBF classifiers, which presented accuracies close to 100% for all subjects independent of the overlap of the segments. ELM and KNN demonstrated a large range in the distributions for more than one subject (i.e., subjects 3, 6, 9, and 12). In this way, these classifiers presented a higher dependence on the subjects compared to RF and SVMRBF.

This first analysis for the subjects took into account that the pattern recognition process uses only the individual data, as the training stage is a calibration process. Then, the second analysis for the subjects was performed, presenting the complete sequence of acquisition trials in the test step. The results of this analysis can be observed in Figure 11. It can be noted that this approach changed the behavior established as observed in the previous analysis, especially in Figure 11a for the data without overlap in segmentation. The SVMRBF, which was one of the classifiers with a high hit rate for a 100% overlap rate, presented accuracies ranging around 20%. Its hit rate only increased with the presence of overlap segments. At the same time, ELM, KNN, and RF classifiers maintained the same pattern observed for the previous analysis, meaning that they are less sensitive to changes in the trails. The Gaussians in SVMRBF could represent a high dependence for each subject, which explains their increase in the hit rates when more samples are present in the distribution by the overlap of the signal.



Figure 11. Hit rates obtained for acquisition trial validation. (a) Accuracies obtained for all subjects considering the variation of signal overlap related to classifiers. (b) Distribution of accuracies for an overlap rate of 12.5% (the best value observed in (a) for each subject and classifier). The bars represent the 25th and 75th percentiles; the whiskers represent approximately 99.3% of the coverage of distribution, and the crosses (+) are the outliers.

Using the best rate in overlap segments (12.5%), Figure 11b shows the obtained distributions for each subject for all classifiers. The ELM classifier presented high dispersion for all subjects, with accuracies ranging from 55% to 85%. The other classifiers presented high values with a concentrated distribution but with a high number of outliers. Some of them reached 20% less than the median in the subjects, especially SVMRBF. KNN and RF presented a similar behavior for all subjects, KNN having a low number of outliers and being close to 100%. However, their distributions did not

present statistical difference (using a Wilcoxon rank sum test, with *p*-value > 0.05), and it established the execution time during the test step. The KNN classifier had an execution time to provide an output for each sample that was 200 times greater than the RF classifier. This was due to the nature of this classifier, also called a "lazy" classifier, because, for each sample, the Euclidean distance is calculated for all of the training data. For this reason, RF had some outliers at up to 87%; however, the accuracies had distributions above 98% for all subjects.

Using the obtained results for the RF classifier for the acquisition trails validation, Figure 12 presents the confusion matrix for the 26 classified gestures. The color bars show the accuracy in each cell, where green cells represent the high hit rates; red cells represent the misclassification samples; and the white cells represent where there was no relevant misclassification (less than 0.3%) values of correspondence. Moreover, the influence of the overlap rate, a parameter that demonstrated a high correlation with the accuracy, is shown. The disruptive windows presented a high dispersion on the hit rates, with an elevated number of false positives and false negatives, until in Figure 12c,d the classes that are more misclassified could be identified.



Figure 12. Confusion matrix for random forest classifier for the classes for all subjects using the validation by the acquisition trials, highlighting the influence of overlap rates of 100% (**a**), 50% (**b**), 25% (**c**), and 12.5% (**d**). The color bar represents the accuracy obtained in each case and the white cells represent values less than 0.3% of misclassification samples.

In general, the best-recognized gestures were "A", "B", "E", "Q", "W", and "Y", with high precision. Regarding gestures that were incorrectly classified, one can note that it occurred where the signals recruited similar movements. For example, this is the case for the gestures from the letters "F" and "T", "C" and "O", "P" and "Q", and "Y" and "Z", that were presented in Figure 1. The reason for this error was recruitment of similar muscle groups (occasioning similar sEMG signals in the channels).

However, it presented accuracy values lower than 0.5%, indicating that the system obtained high hit rates for each sign.

Lastly, the data for men and women were separated to analyze if there was difference between them in the classification. In the average for the best classifier and feature set (G9 and RF) for the observed segmentation parameters (1.75 s of window length and 12.5% of overlap rate), men reached accuracies of around 0.9881 and women 0.9906. The Wilcoxon test (p < 0.05) was used to verify if the results allow the same distribution. For the first moment, the test did not find a significant difference between them, the *p*-value being higher (the *p*-value obtained was 0.461). However, due to the number of subjects, 10 men and two women, and close accuracy, it is not possible to conclude on a prevalence of performance between genders.

4.5. Comparison with Related Works

Aiming to compare the obtained results with those presented in the literature, Table 5 summarizes the main studies involving sEMG processing for SLRs that are correlated with the approach mentioned in this paper.

Before the discussion, it is necessary to briefly mention the obtained results. Regarding signal segmentation, window length and overlap rate parameters were evaluated. The overlapped segmentation was the parameter that most affected the accuracy of the classifiers; the disruptive window segmentation resulted in low accuracies. In the feature extraction stage, feature sets based on the literature were evaluated. In this work, a feature set (G9) formed by MFL, MNP, TTP, and RMS was proposed, which exhibited similar performance and reduced processing time for the literature feature sets. Concerning the classifiers, we began our investigation with nine classifiers and four of them exhibited performance above the average: ELM, KNN, RF, and SVMRBF. During the validation process in the analysis for each subject, RF demonstrated robustness with high accuracies and a short classification time for each sample. Overall, the pattern recognition process reached 99% for the RF classifier, with feature set G9, a window length of 1.75 s, and an overlap rate of 12.5%.

Comparing the number of signs, a closed set of gestures was considered, and comparing with the works that had Libras recognition as an objective for sEMG, 26 signs were used (more than the 20 used in [38,39]). Besides that, signals from 12 subjects (both male and female) were evaluated, higher than the similar works that explored between three and eight subjects. To contribute the findings regarding the individual classification to a generalist process, the validation stages were tested separately for all dataset, for each subject, and for the acquisition trials. The accuracies remained similar in all these processes.

Regarding the number of channels and acquisition mode, it is notable that the armband approach is a trend for gesture recognition, which has appeared also in SLR systems. Differently from fixed placement of the electrodes, the device is positioned with a single orientation, easing the process for people that are not specialist in anatomy or muscle localization on the forearm. Another proposition is the use of inertial unities (such as accelerometers) to aid the recognition process. In this work, only sEMG signals were considered, verifying this contribution in this process. It is right to affirm that inertial sensors can help this kind of application, but it is necessary to validate how much the sEMG supports Libras recognition.

Work	SLR	Signals/ Subjects	sEMG Channels and/Other Sensors	Window Length/ Overlapping	Features	Classifier	Results
[25]	ASL	9/-	2	512 samples/Y	IAV, ZC, DAMV, AR, CEPS, Mean frequency	DA	97%
[10]	GSL	60/3	5/Accelerometer	-/-	Sample Entropy	DA	93%
[24]	CSL	72/2	5/Accelerometer	100 ms/N	MAV, AR	DT, HMM	98%
[29]	CSL	223/5	4/Accelerometer	64 points/Y	MAV, AR	DTW, HMM	92% to 96%
[28]	ASL	40/4	4/Accelerometer Gyroscope Magnetometer	128 ms/N	MAV, AR, HIST, RMS, Reflection Coefficients, VAREMG, WAMP, MDF, Modify MNF	NB, KNN, DT, SVM	96%
[27]	CSL	121/5	4/Accelerometer	128 ms/Y	MAV, AR, Mean, VAREMG, Linear Prediction Coefficients	RF, HMM	98%
[11]	ASL	80/4	4/ Accelerometer	128 ms/N	Reflection Coefficients, VAREMG, WAMP, MDF, Modify MNF	NB, KNN, DT, SVM	96%
[38]	Libras	20/-	8 (Myo armband) 8 (Myo armband)/	-/-	Mean	SVM	41%
[26]	KSL	30/6	Accelerometer Gyroscope Magnetometer	200 ms/N	Raw signal	CNN	98%
[23]	CSL	18/8	5/ Accelerometer	176 ms/Y 50-350 ms/Y	MAV, ZC, SSC, and WL	LDA	91%
[39]	Libras	20/1	8 (Myo armband)	750 ms/N	IEMG, MAV, RMS, LOGDEC, ZC, SSC, MNF, PKF, MNP, SM0, SM1	MLP	81%
This Work	Libras	26/12	8 (Myo armband)	Window Length: 0.25 to 2.25 s/ Without overlap, 50%, 25%, and 12.5%	AK4, CEP5, DASDV, HIS1, IEMG, LOGDEC, LS, MAV, MAV1, MAV2, MFL, MSR, MYOP, RMS, SampEn, SSC, TM3, TM4, TM5, VAREMG, VORDER, WAMP, WL, ZC, FR, MDF, MNF, PKF, SM1, SM2, SM3, TTP	KNN, NB, LDA, QDA, RF, ELM, MLP, SVMLin, SVMRBF	99%

 Table 5. Sign Language recognition (SLR) systems with sEMG as input and their main characteristics.

In segmentation, a great impact on the hit rates was found, or in other words on the influence of the SLR. These parameters were not examined appropriately in the SLRs works. The works in Table 5 had a window length of between 200 and 750 ms, reaching accuracies between 80% and 98%. These values are lower than the values used in this work, where the best values were achieved for 1 and 2.25 s. We also searched for a generalist model (with data from more subjects), which was evaluated for more than one value of window length. Nevertheless, it is important to mention that for overlap rates of 25% and 12.5%, window lengths of 0.25 and 0.5 s reached accuracies close to 90% for the RF and KNN classifiers (as demonstrated in Figure 7). Moreover, overlapping signals may increase the number and variability of instances for the classification system training, which should be even more relevant for approaches that demand large datasets. Classifying in smaller steps with overlapping has a similar meaning of increasing the system resolution, as more data samples are processed in a given epoch of the time series. However, making the window step small may increase the processing time, and making the window size smaller would reduce the information within it. In this way, a trade-off should be reached between being feasible for real-time operation and having time resolution and classification accuracy.

Concerning the features, similar works applied some features and feature sets; however, they did not present their influence on the classification process. Both single features and feature sets approaches were evaluated in this work and the G9 set was considered suitable to be used for the armband approach in Libras recognition. Some evidence and hypotheses explain why the G9 set presented these values of the hit rate. RMS is closely related to MAV (features that are present in a large number of works in Table 3), which is a time-domain feature that provides amplitude information. MFL is a feature that has presented high accuracies for systems with higher sampling rates than the MyoTM device [55]. TTP and MNP are frequency-domain features. Moreover, these feature sets presented a lower computational cost compared to similar literature feature sets, being more suitable to be used in embedded systems.

A large number of different classifiers was evaluated in this work, in contrast to the works in Table 5 that applied around four classifiers. From the nine classifiers, four exhibited high accuracies for the alphabet in Libras: RF, KNN, SVMRBF, and ELM, in this order of accuracy. As in [27], the random forests exhibited the highest accuracy and was more suitable for an embedded application. Moreover, in this work, two methods of validation were tested: one based on the single subject test and another on an acquisition trial, the last being a generalist way to evaluate the signs.

5. Conclusions

This paper presented the recognition of Brazilian Sign Language and the influence of signal processing parameters during the classification process: signal segmentation, feature extraction, and pattern recognition. The sEMG signals were collected using an armband approach using the wearable device MyoTM. Data from 12 subjects were acquired and the 26 letters of the Brazilian Sign Language were classified. Only sEMG signals were used in this approach, verifying the contribution of this signal to this task.

The main purpose of this paper was to demonstrate how the data processing steps and parameters influence the SLR process, detailing the contribution of each stage to the accuracy. It was found that the segmentation was the parameter with more influence in the pattern recognition process, and especially the overlap rate. This can be explained due to the increase in the useful information and the instances for training, and it makes it suitable for embedded applications (if used together with the RF classifier). In overlapping, it is more important to consider for future developments time-dependent models and post-processing techniques, such as majority vote, to guarantee high accuracies.

The features were found to be a second factor of importance, mainly in the evaluation of the feature sets. It was demonstrated that features from both the time and frequency domains need to be used and it is possible to use a feature set with similar hit rates in the literature with low processing time,

such as in the case of G9. Future works can explore the influence of the features on other segmentation parameters, detecting if there are significant changes.

In classification, RF was robust and presented high values in all the tests. Overall, a high accuracy (reaching 99%) was found. To test the robustness of this method, the separation of the test set for the acquisition trials was proposed and the system kept similar hit rates. The classes with more than 0.03% (indicated in Figure 12d) showed that the gestures with more confusion were the signs with similar movements, which represent similar muscle activations. Future works can explore the use of classifiers that are not dependent on handcraft feature selection, like CNN, deep learning techniques, and classifiers with time interval information, such as HMM.

The single sEMG armband approach for signal acquisition proved capable of recognizing signs from Libras. For future works, the database could be increased in terms of both number of gestures and signs, and the acquisition parameters could be analyzed, for example the number of channels; in addition, comparisons could be made with the fixed electrode approach and between men and women.

Author Contributions: Conceptualization, J.J.A.M.J., S.L.S.J., and S.F.P.; methodology, J.J.A.M.J., D.P.C., and M.L.B.F.; software, J.J.A.M.J., M.L.B.F., and F.A.F.; validation, J.J.A.M.J., F.A.F., and D.P.C., Y.Y.; formal analysis, J.J.A.M.J., S.L.S.J.; investigation, J.J.A.M.J.; resources, J.J.A.M.J.; data curation, J.J.A.M.J.; writing—original draft preparation, J.J.A.M.J., M.L.B.F., D.P.C., and F.A.F.; writing—review and editing, S.L.S.J. and S.F.P.; visualization, J.J.A.M.J.; supervision, S.F.P. and S.L.S.J.; project administration, S.F.P. and S.L.S.J.; funding acquisition, S.F.P. All authors have read and agreed to the published version of the manuscript.

Funding: This study was financed in part by the Coordenação de Aperfeiçoamento de Pessoal de Nível Superior-Brasil (CAPES)-Finance Code 001, of Conselho Nacional de Desenvolvimento Científico e Tecnológico (CNPq).

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

The following abbreviations are used in this manuscript:

ADC	Analog-Digital Converter
ANN	Artificial Neural Network
AR	Autoregressive Coefficients
AR4	4-th order Autoregressive Coefficients
ASL	American Sign Language
bpm	Beats per minute
CEPS	Cepstral Coefficients
CNN	Convolution Neural Network
CSL	Chinese Sign Language
DA	Discriminant Analysis
DAMV	Difference of Absolute Mean Value
DASDV	Difference Absolute of Standard Deviation Value
DT	Decision Tree
DTW	Dynamic Time Warping
ELM	Extreme Learning Machine
FR	Frequency Rate
GSL	Greek Sign Language
HIST	Histogram
HMM	Hidden Markov Model
IAV	Integrated Absolute Value
IEMG	Integral of EMG
IMU	Inertial Measurement Unity
ISL	Indonesian Sign Language
KNN	k-Nearest Neighbor
KSL	Korean Sign Language
LDA	Linear Discriminant Analysis
Libras	Brazilian Sign Language

Log detector
L-Scale
Mean Average Value
Median Frequency
Multi-Layer Perceptron
Mean Frequency
Mean Power
Myopulse percentage rate
Naïve Bayes
Peak Frequency
Quadratic Discriminant Analysis
Random Forest
Root Mean Square
Sample Entropy
surface Electromyography
Sign Language
Sign Language Recognition
Spectral Moment
Sign Slope Crossing
Support Vector Machine
Absolute value of temporal moment
Thai Sign Language
Total power
Variance of EMG
<i>v</i> -Order
Willison Amplitude
Waveform Length
Zero Crossing

Appendix A

To obtain the acquired raw sEMG signal, please contact the correspondence e-mail: josej@alunos.utfpr.edu.br.

References

- Cheok, M.J.; Omar, Z.; Jaward, M.H. A review of hand gesture and sign language recognition techniques. *Int. J. Mach. Learn. Cyber.* 2017, 1–23. [CrossRef]
- Abdallah, E.E.; Fayyoumi, E. Assistive Technology for Deaf People Based on Android Platform. *Procedia* Comput. Sci. 2016, 94, 295–301. [CrossRef]
- 3. Courtin, C. The impact of sign language on the cognitive development of deaf children: The case of theories of mind. *J. Deaf Stud. Deaf Educ.* **2000**, *5*, 266–276. [CrossRef] [PubMed]
- Hommes, R.E.; Borash, A.I.; Hartwig, K.; DeGracia, D. American Sign Language Interpreters Perceptions of Barriers to Healthcare Communication in Deaf and Hard of Hearing Patients. *J. Community Health* 2018, 43, 956–961. [CrossRef] [PubMed]
- 5. Azar, S.G.; Seyedarabi, H. Trajectory-based recognition of dynamic Persian sign language using hidden Markov model. *Comput. Speech Lang.* **2020**, *61*, 101053. [CrossRef]
- 6. Almasre, M.A.; Al-Nuaim, H. A comparison of Arabic sign language dynamic gesture recognition models. *Heliyon* **2020**, *6*, e03554. [CrossRef]
- 7. Al-Ahdal, M.E.; Nooritawati, M.T. Review in Sign Language Recognition Systems. In Proceedings of the 2012 IEEE Symposium on Computers Informatics (ISCI), Penang, Malaysia, 18–20 March 2012; pp. 52–57.
- Riillo, F.; Quitadamo, L.R.; Cavrini, F.; Saggio, G.; Pinto, C.A.; Pasto, N.C.; Sbernini, L.; Gruppioni, E. Evaluating the influence of subject-related variables on EMG-based hand gesture classification. In Proceedings of the 2014 IEEE International Symposium on Medical Measurements and Applications (MeMeA), Lisbon, Portugal, 11–12 June 2014; pp. 1–5.

- 9. Yavuz, E.; Eyupoglu, C. A cepstrum analysis-based classification method for hand movement surface EMG signals. *Med. Biol. Eng. Comput.* **2019**, *57*, 2179–2201. [CrossRef]
- 10. Kosmidou, V.E.; Hadjileontiadis, L.J. Sign Language Recognition Using Intrinsic-Mode Sample Entropy on sEMG and Accelerometer Data. *IEEE Trans. Biomed. Eng.* **2009**, *56*, 2879–2890. [CrossRef]
- 11. Wu, J.; Sun, L.; Jafari, R. A Wearable System for Recognizing American Sign Language in Real-Time Using IMU and Surface EMG Sensors. *IEEE J. Biomed. Health Inform.* **2016**, *20*, 1281–1290. [CrossRef]
- 12. Jaramillo-Yánez, A.; Benalcázar, M.E.; Mena-Maldonado, E. Real-Time Hand Gesture Recognition Using Surface Electromyography and Machine Learning: A Systematic Literature Review. *Sensors* 2020, 20, 2467. [CrossRef]
- 13. Criswell, E. *Cram's Introduction to Surface Electromyography;* Jones and Bartlett Publishers: Sudbury, MA, USA, 2011.
- 14. Raez, M.B.I.; Hussain, M.S.; Mohd-Yasin, F. Techniques of EMG signal analysis: Detection, processing, classification and applications. *Biol. Proced. Online* **2006**, *8*, 11–35. [CrossRef] [PubMed]
- Li, Y.; Chen, X.; Zhang, X.; Wang, K.; Wang, Z.J. A Sign-Component-Based Framework for Chinese Sign Language Recognition Using Accelerometer and sEMG Data. *IEEE Trans. Biomed. Eng.* 2012, 59, 2695–2704. [CrossRef] [PubMed]
- Abbaspour, S.; Lindén, M.; Gholamhosseini, H.; Naber, A.; Ortiz-Catalan, M. Evaluation of surface EMG-based recognition algorithms for decoding hand movements. *Med. Biol. Eng. Comput.* 2020, *58*, 83–100. [CrossRef] [PubMed]
- 17. Oskoei, M.A.; Hu, H. Support Vector Machine-Based Classification Scheme for Myoelectric Control Applied to Upper Limb. *IEEE Trans. Biomed. Eng.* **2008**, *55*, 1956–1965. [CrossRef]
- Saikia, A.; Kakoty, N.M.; Phukan, N.; Balakrishnan, M.; Sahai, N.; Paul, S.; Bhatia, D. Combination of EMG Features and Stability Index for Finger Movements Recognition. *Procedia Comput. Sci.* 2018, 133, 92–98. [CrossRef]
- Phinyomark, A.; Quaine, F.; Charbonnier, S.; Serviere, C.; Tarpin-Bernard, F.; Laurillau, Y. EMG feature evaluation for improving myoelectric pattern recognition robustness. *Expert Syst. Appl.* 2013, 40, 4832–4840. [CrossRef]
- Barioul, R.; Fakhfakh, S.; Derbel, H.; Kanoun, O. Evaluation of EMG Signal Time Domain Features for Hand Gesture Distinction. In Proceedings of the 2019 16th International Multi-Conference on Systems, Signals Devices (SSD), Istanbul, Turkey, 21–24 March 2019; pp. 489–493.
- 21. Shroffe, E.H.; Manimegalai, P. Hand Gesture Recognition based on EMG signals using ANN. *Int. J. Comput. Appl.* **2013**, *2*, 31–39.
- 22. Derr, C.; Sahin, F. Signer-independent classification of American sign language word signs using surface EMG. In Proceedings of the 2017 IEEE International Conference on Systems, Man, and Cybernetics (SMC), Banff, Canada, 5–8 October 2017; pp. 665–670.
- Zhuang, Y.; Lv, B.; Sheng, X.; Zhu, X. Towards Chinese sign language recognition using surface electromyography and accelerometers. In Proceedings of the 2017 24th International Conference on Mechatronics and Machine Vision in Practice (M2VIP), Auckland, New Zealand, 21–23 November 2017; pp. 1–5.
- 24. Zhang, X.; Chen, X.; Li, Y.; Lantz, V.; Wang, K.; Yang, J. A Framework for Hand Gesture Recognition Based on Accelerometer and EMG Sensors. *IEEE Trans. Syst. Man Cybern.—Part A Syst. Hum.* **2011**, *41*, 1064–1076. [CrossRef]
- 25. Kosmidou, V.E.; Hadjileontiadis, L.J.; Panas, S.M. Evaluation of surface EMG features for the recognition of American Sign Language gestures. In Proceedings of the 2006 International Conference of the IEEE Engineering in Medicine and Biology Society, New York, NY, USA, 30 August–3 September 2006; pp. 6197–6200.
- 26. Shin, S.; Baek, Y.; Lee, J.; Eun, Y.; Son, S.H. Korean sign language recognition using EMG and IMU sensors based on group-dependent NN models. In Proceedings of the 2017 IEEE Symposium Series on Computational Intelligence (SSCI), Honolulu, HA, USA, 27 November–1 December 2017; pp. 1–7.
- 27. Su, R.; Chen, X.; Cao, S.; Zhang, X. Random Forest-Based Recognition of Isolated Sign Language Subwords Using Data from Accelerometers and Surface Electromyographic Sensors. *Sensors* **2016**, *16*, 100. [CrossRef]

- 28. Wu, J.; Tian, Z.; Sun, L.; Estevez, L.; Jafari, R. Real-time American Sign Language Recognition using wrist-worn motion and surface EMG sensors. In Proceedings of the 2015 IEEE 12th International Conference on Wearable and Implantable Body Sensor Networks (BSN), Cambridge, MA, USA, 9–12 June 2015; pp. 1–6.
- 29. Cheng, J.; Chen, X.; Liu, A.; Peng, H. A Novel Phonology- and Radical-Coded Chinese Sign Language Recognition Framework Using Accelerometer and Surface Electromyography Sensors. *Sensors* **2015**, *15*, 23303–23324. [CrossRef]
- Amatanon, V.; Chanhang, S.; Naiyanetr, P.; Thongpang, S. Sign language-Thai alphabet conversion based on Electromyogram (EMG). In Proceedings of the The 7th 2014 Biomedical Engineering International Conference, Fukuoka, Japan, 26–28 November 2014; pp. 1–4.
- 31. Khamid; Wibawa, A.D.; Sumpeno, S. Gesture Recognition for Indonesian Sign Language Systems (ISLS) Using Multimodal Sensor Leap Motion and Myo Armband Controllers Based-on Naïve Bayes Classifier. In Proceedings of the 2017 International Conference on Soft Computing, Intelligent System and Information Technology (ICSIIT), Denpasar, Indonesia, 26–29 September 2017; pp. 1–6.
- 32. Rahagiyanto, A.; Basuki, A.; Sigit, R. Moment Invariant Features Extraction for Hand Gesture Recognition of Sign Language based on SIBI. *EMITTER Int. J. Eng. Technol.* **2017**, *5*, 119–138. [CrossRef]
- Bastos, I.L.O.; Angelo, M.F.; Loula, A.C. Recognition of Static Gestures Applied to Brazilian Sign Language (Libras). In Proceedings of the 2015 28th SIBGRAPI Conference on Graphics, Patterns and Images, Salvador, Brazil, 26–29 August 2015; pp. 305–312.
- Costa Filho, C.F.F.; Souza, R.S.D.; Santos, J.R.D.; Santos, B.L.D.; Costa, M.G.F.; Costa Filho, C.F.F.; Souza, R.S.D.; Santos, J.R.D.; Santos, B.L.D.; Costa, M.G.F. A fully automatic method for recognizing hand configurations of Brazilian sign language. *Res. Biomed. Eng.* 2017, *33*, 78–89. [CrossRef]
- Amaral, L.; Júnior, G.L.N.; Vieira, T.; Vieira, T. Evaluating Deep Models for Dynamic Brazilian Sign Language Recognition. In Proceedings of the Progress in Pattern Recognition, Image Analysis, Computer Vision, and Applications; Vera-Rodriguez, R., Fierrez, J., Morales, A., Eds.; Springer International Publishing: Cham, Switzerland, 2019; pp. 930–937.
- Freitas, M.L.B.; Mendes Junior, J.J.A.; Campos, D.P.; Stevan, S.L., Jr. Hand gestures classification using multichannel sEMG armband. In Proceedings of the Anais do XXVI Congresso Brasileiro de Engenharia Biomédica, Armação dos Búzios, Brazil, 21–25 October 2018.
- 37. Côté-Allard, U.; Gagnon-Turcotte, G.; Laviolette, F.; Gosselin, B. A Low-Cost, Wireless, 3-D-Printed Custom Armband for sEMG Hand Gesture Recognition. *Sensors* **2019**, *19*, 2811. [CrossRef] [PubMed]
- Abreu, J.G.; Teixeira, J.M.; Figueiredo, L.S.; Teichrieb, V. Evaluating Sign Language Recognition Using the Myo Armband. In Proceedings of the 2016 XVIII Symposium on Virtual and Augmented Reality (SVR), Gramado, Brazil, 21–24 June 2016; pp. 64–70.
- Mendes Junior, J.J.A.; Freitas, M.L.B.; Stevan, S.L.; Pichorim, S.F. Recognition of Libras Static Alphabet with MyoTM and Multi-Layer Perceptron. In XXVI Brazilian Congress on Biomedical Engineering; Costa-Felix, R., Machado, J.C., Alvarenga, A.V., Eds.; Springer: Singapore, 2019; pp. 413–419.
- Asogbon, M.G.; Samuel, O.W.; Geng, Y.; Chen, S.; Mzurikwao, D.; Fang, P.; Li, G. Effect of Window Conditioning Parameters on the Classification Performance and Stability of EMG-Based Feature Extraction Methods. In Proceedings of the 2018 IEEE International Conference on Cyborg and Bionic Systems (CBS), Shenzhen, China, 25–27 October 2018; pp. 576–580.
- 41. Vaitkevičius, A.; Taroza, M.; Blažauskas, T.; Damaševičius, R.; Maskeliūnas, R.; Woźniak, M. Recognition of American Sign Language Gestures in a Virtual Reality Using Leap Motion. *Appl. Sci.* **2019**, *9*, 445. [CrossRef]
- 42. Saggio, G.; Cavallo, P.; Ricci, M.; Errico, V.; Zea, J.; Benalcázar, M.E. Sign Language Recognition Using Wearable Electronics: Implementing k-Nearest Neighbors with Dynamic Time Warping and Convolutional Neural Network Algorithms. *Sensors* **2020**, *20*, 3879. [CrossRef]
- 43. Chong, T.-W.; Kim, B.-J. American Sign Language Recognition System Using Wearable Sensors with Deep Learning Approach. *J. Korea Inst. Electron. Commun. Sci.* **2020**, *15*, 291–298. [CrossRef]
- 44. Rastgoo, R.; Kiani, K.; Escalera, S. Hand sign language recognition using multi-view hand skeleton. *Expert Syst. Appl.* **2020**, *150*, 113336. [CrossRef]
- 45. Xiao, Q.; Qin, M.; Yin, Y. Skeleton-based Chinese sign language recognition and generation for bidirectional communication between deaf and hearing people. *Neural Netw.* **2020**, *125*, 41–55. [CrossRef]
- 46. Chen, L.; Fu, J.; Wu, Y.; Li, H.; Zheng, B. Hand Gesture Recognition Using Compact CNN via Surface Electromyography Signals. *Sensors* **2020**, *20*, 672. [CrossRef]

- 47. Guo, H.; Sung, Y. Movement Estimation Using Soft Sensors Based on Bi-LSTM and Two-Layer LSTM for Human Motion Capture. *Sensors* **2020**, *20*, 1801. [CrossRef]
- 48. Visconti, P.; Gaetani, F.; Aravind, G.; Zappatore; Primiceri, P. Technical Features and Functionalities of Myo Armband: An Overview on Related Literature and Advanced Applications of Myoelectric Armbands Mainly Focused on Arm Prostheses. *Int. J. Smart Sens. Intell. Syst.* **2018**, *11*, 1–25. [CrossRef]
- 49. Kamen, G.; Gabriel, D. Essentials of Electromyography: Human Kinetics. Available online: https://us. humankinetics.com/products/essentials-of-electromyography (accessed on 2 August 2020).
- Farrell, T.R.; Weir, R.F. A Comparison of the Effects of Electrode Implantation and Targeting on Pattern Classification Accuracy for Prosthesis Control. *IEEE Trans. Biomed. Eng.* 2008, 55, 2198–2211. [CrossRef] [PubMed]
- 51. Hakonen, M.; Piitulainen, H.; Visala, A. Current state of digital signal processing in myoelectric interfaces and related applications. *Biomed. Signal Process. Control* **2015**, *18*, 334–359. [CrossRef]
- 52. Al-Mulla, M.R.; Sepulveda, F.; Colley, M. sEMG Techniques to Detect and Predict Localised Muscle Fatigue. *EMG Methods Eval. Muscle Nerve Funct.* **2012**. [CrossRef]
- 53. Mendes Junior, J.J.A.; Freitas, M.L.B.; Siqueira, H.V.; Lazzaretti, A.E.; Pichorim, S.F.; Stevan, S.L. Feature selection and dimensionality reduction: An extensive comparison in hand gesture classification by sEMG in eight channels armband approach. *Biomed. Signal Process. Control* **2020**, *59*, 101920. [CrossRef]
- 54. Hudgins, B.; Parker, P.; Scott, R.N. A new strategy for multifunction myoelectric control. *IEEE Trans. Biomed. Eng.* **1993**, *40*, 82–94. [CrossRef]
- 55. Phinyomark, A.; N. Khushaba, R.; Scheme, E. Feature Extraction and Selection for Myoelectric Control Based on Wearable EMG Sensors. *Sensors* **2018**, *18*, 1615. [CrossRef]
- 56. Liu, Y.; Huang, H.; Weng, C. Recognition of Electromyographic Signals Using Cascaded Kernel Learning Machine. *IEEE/ASME Trans. Mechatron.* **2007**, *12*, 253–264. [CrossRef]
- 57. Tkach, D.; Huang, H.; Kuiken, T.A. Study of stability of time-domain features for electromyographic pattern recognition. *J. Neuroeng. Rehabil.* **2010**, *7*, 21. [CrossRef]
- 58. Phinyomark, A.; Phukpattaranont, P.; Limsakul, C. Feature reduction and selection for EMG signal classification. *Expert Syst. Appl.* **2012**, *39*, 7420–7431. [CrossRef]
- 59. Zhang, X.; Zhou, P. Sample entropy analysis of surface EMG for improved muscle activity onset detection against spurious background spikes. *J. Electromyogr. Kinesiol.* **2012**, *22*, 901–907. [CrossRef] [PubMed]
- 60. Kaczmarek, P.; Mańkowski, T.; Tomczyński, J. putEMG—A Surface Electromyography Hand Gesture Recognition Dataset. *Sensors* **2019**, *19*, 3548. [CrossRef]
- 61. Wu, Y.; Liang, S.; Zhang, L.; Chai, Z.; Cao, C.; Wang, S. Gesture recognition method based on a single-channel sEMG envelope signal. *J. Wirel. Commun. Network.* **2018**, 35. [CrossRef]
- Chang, C.C.; Lin, C.J. LIBSVM: A library for support vector machines. ACM Trans. Intell. Syst. Technol. 2011, 2, 1–27. [CrossRef]
- 63. Slutter, M.W.J. Creating a Feedback System with the Myo Armband, for Home Training for Frail Older Adults. Bachelor Thesis, University of Twente, Enschede, Netherlands, 2017.
- 64. Pizzolato, S.; Tagliapietra, L.; Cognolato, M.; Reggiani, M.; Müller, H.; Atzori, M. Comparison of six electromyography acquisition setups on hand movement classification tasks. *PLoS ONE* **2017**, *12*, e0186132. [CrossRef] [PubMed]
- 65. Kumar, D.K.; Jelfs, B.; Sui, X.; Arjunan, S.P. Prosthetic hand control: A multidisciplinary review to identify strengths, shortcomings, and the future. *Biomed. Signal Process. Control* **2019**, *53*, 101588. [CrossRef]
- 66. Smith, L.H.; Hargrove, L.J.; Lock, B.A.; Kuiken, T.A. Determining the optimal window length for pattern recognition-based myoelectric control: Balancing the competing effects of classification error and controller delay. *IEEE Trans. Neural. Syst. Rehabil. Eng.* **2011**, *19*, 186–192. [CrossRef]
- 67. Demšar, J. Statistical Comparisons of Classifiers over Multiple Data Sets. J. Mach. Learn. Res. 2006, 7, 1–30.



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).